Title: A comparison of patient recall of smoking cessation advice with advice recorded in electronic medical records

Authors:

Lisa Szatkowski (lisa.szatkowski@nottingham.ac.uk)
Ann McNeill (ann.mcneill@nottingham.ac.uk)
Sarah Lewis (sarah.lewis@nottingham.ac.uk)
Tim Coleman (tim.coleman@nottingham.ac.uk)

Version: 3 Date: 3 May 2011

Author's response to reviews:

3 May 2011

Dear Dr Samet

Thank you for accepting our article for publication in BMC Public Health. Please find attached a revised version of our manuscript, in which we have:

- provided each author's affiliation in full.
- altered the section headings as requested.
- removed all highlighting and tracking from the manuscript.
- removed 'insert figure here' notes and instead provided a figure legend after the references.
- checked our manuscript for any typographical errors and made any final corrections.

With best wishes,

Lisa Szatkowski